Author's response to reviews

Title: Herpes encephalitis presenting as mild aphasia: Case Report

Authors:

Omar A. Khan (omar.khan@vtmednet.org)
Allan Ramsay (allan.ramsay@vtmednet.org)

Version: 5 Date: 21 March 2006

Author's response to reviews:

Dear Editor:

We have made the formatting changes requested:

--Title & Abstract - Please ensure these are identical in the manuscript and in the submission system.

Done.

--Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Done.

--Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Case Presentation; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any). The Background should provide an introduction to the subject, and an account of why a new method is needed.

Done.

Sincerely yours,

Omar A. Khan
Allan Ramsay